Author's response to reviews

Title: When battery exhaustion lets the lame walk: A case report on the importance of long-term stimulator monitoring in deep brain stimulation

Authors:

Martin Sommer (msommer@gwdg.de)
Elisabeth M Stiksrud (e.stiksrud@gmx.de)
Kajetan von Eckardstein (kajetan.voneckardstein@med.uni-goettingen.de)
Veit Rohde (veit.rohde@med.uni-goettingen.de)
Walter Paulus (wpaulus@med.uni-goettingen.de)

Version: 2 Date: 29 October 2014

Author's response to reviews:

Dear Jhonell De Los Santos,

thank you for your mail. I have included line and page numbers, as well as an authors' contribution section, and a care checklist indicating the line numbers where the points have been addressed.

Yours sincerely,

Martin Sommer